Author's response to reviews

Title: A cluster randomised controlled trial evaluating the effectiveness of eHealth-supported patient recruitment in primary care research: The TRANSFoRm study protocol

Authors:

Nikolaos Mastellos (n.mastellos@imperial.ac.uk)
Anna Andreasson (anna.andreasson@ki.se)
Kit Huckvale (c.huckvale@imperial.ac.uk)
Mark Larsen (m.larsen@imperial.ac.uk)
Vasa Curcin (vasa.curcin@kcl.ac.uk)
Josip Car (josip.car@imperial.ac.uk)
Lars Agreus (lars.agreus@ki.se)
Brendan Delaney (brendan.delaney@kcl.ac.uk)

Version: 2 Date: 7 January 2015

Author's response to reviews:

Revised manuscript to fully comply with the CONSORT recommendations for the reporting of cluster randomised controlled trials